Author’s response to reviews

Title: Computational analysis of receptor tyrosine kinase inhibitors and cancer metabolism: Implications for treatment and discovery of potential therapeutic signatures

Authors:

Jian Li (simon_li82@hotmail.com)

Kathrin Halfter (halfter@ibe.med.uni-muenchen.de)

Mengying Zhang (mzhang@ibe.med.uni-muenchen.de)

Christian Saad (saad@informatik.uni-augsburg.de)

Kai Xu (godocoto@163.com)

Bernhard Bauer (Bauer@informatik.uni-augsburg.de)

Yijiang Huang (yijiang0116@yahoo.com)

Shi Lei (leishinsh@163.com)

Ulrich Mansmann (mansman@ibe.med.uni-munchen.de)

Version: 2 Date: 04 Jun 2019

Author’s response to reviews:

BCAN-D-18-02898: Computational analysis of receptor tyrosine kinase inhibitors and cancer metabolism: Implications for treatment and discovery of potential therapeutic signatures

Dear Editor,

Thank you for your response and reviewers comments. We have modified this version of manuscript in response to your comments. Below please find a point-by-point response to your comments.

We also checked that the new version of manuscript does adhere to the editorial requirements and hope that this new version is sufficient for the approval of the article for publication in BMC Cancer.

Your sincerely,
Dr. Jian Li

Editor Comments:

1. Please ensure that sufficient information is provided to the reader to be able to access the relevant data from:

http://www.broadinstitute.org/ccle.

Answer: we provide the Gene Expression Omnibus (GEO) with accession number GSE36139 in the section of Availability of data and materials (Page 10) to ensure the availability of data.

2. Please clarify whether any of the authors have competing interests.

Answer: we declare that there is no competing interests among the authors and affiliations in the section of Competing interest (Page 10).

3. Figure titles (max 15 words) and legends (max 300 words) should be provided in the main manuscript after the References.

Answer: Figure titles and legends have been provided in the new version of manuscript after the References.

4. At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Answer: the proofread manuscript is provided as a single, final and clean version.